

Housing & Society

Journal of the Housing Education and Research Association

Reviewer's Comments for Manuscript Review

Reviewer #:

Manuscript #:

Date:

Review: 1st _____ 2nd _____

Title:

Thank you for reviewing this manuscript. If for some reason you cannot complete the review, please notify me. Please use this form for your recommendations concerning the manuscript, its publication status, and remarks. Comments to the editor will be confidential and will not be sent to the author or other reviewers. Please return the completed forms by the date on your e-mail notification.

Please email or send to:

Becky Yust
University of Minnesota
College of Design
240 McNeal Hall, 1985 Buford Avenue
St. Paul, MN 55108

Please keep the manuscript and all correspondence until you receive word from the editor concerning its final acceptance or rejection. At that time, all copies should be discarded. Please remember that any unpublished manuscripts are the property of the author(s) and cannot be quoted, copied, or circulated without the author's permission. Reviews are blind and author and reviewer identities are anonymous.

Comments to the editor:

Please check manuscript recommendation:

_____ The manuscript should be published and is acceptable after routine editing and need not be returned to the author.

_____ The manuscript should be published but requires minor revision. The author should revise.

_____ The manuscript requires major revision. However, it may be published after revisions and resubmission for further review.

Do you wish to review the revised manuscript?

___ Yes ___ No

_____ The manuscript should not be published.

Manuscript Review Guidelines

Please use this form for reviewing the manuscript. Additional remarks may be included on a separate sheet or on the back of this form. Comments will be given to the primary author.

Reviewer #:

Manuscript #:

Criteria	Excellent					Poor	Comments
Significant subject for HERA	5	4	3	2	1		
Sound theory/conceptual framework	5	4	3	2	1		
Appropriate research design	5	4	3	2	1		
Appropriate data analysis	5	4	3	2	1		
Accurate interpretation of results	5	4	3	2	1		
Appropriate conclusions	5	4	3	2	1		
Relevant implications	5	4	3	2	1		
Up-to-date and relevant references	5	4	3	2	1		
Tables/figures that increase interest	5	4	3	2	1		
Clear writing style	5	4	3	2	1		
Logical organization	5	4	3	2	1		
Observed submission criteria	5	4	3	2	1		

Comments to the author(s):